Author’s response to reviews

Title: Reduced serum levels of pro-inflammatory chemokines in Fragile X Syndrome

Authors:

Anke Van Dijck (Anke.VanDijck@uantwerpen.be)

Susana Barbosa (barbosa@ipmc.cnrs.fr)

Patricia Bermudez-Martin (bermudez@ipmc.cnrs.fr)

Olfa Khalfallah (khalfallah@ipmc.cnrs.fr)

Cyprien Gilet (gilet@i3s.unice.fr)

Emanuela Martinuzzi (martinuzzi@ipmc.cnrs.fr)

Ellen Elinck (Ellen.Elinck@uantwerpen.be)

R Frank Kooy (frank.kooy@uantwerpen.be)

Nicolas Glaichenhaus (glaichenhaus@ipmc.cnrs.fr)

Laetitia Davidovic (davidovic@ipmc.cnrs.fr)

Version: 4 Date: 26 Mar 2020

Author’s response to reviews:

Dear Editor,

We are submitting a novel version of our initial manuscript (MS# NURL-D-19-00686) entitled “Reduced serum levels of pro-inflammatory chemokines in Fragile X Syndrome” for consideration in BMC Neurology.

The last requested modifications are detailed in the point-by-point response, appended to this letter.

Having now complied with your requests, we hope that this manuscript would now be suitable for publication in BMC Neurology.

I remain, yours faithfully,
Laetitia Davidovic, PhD

Answers to requests for manuscript modifications before final acceptation

1. Please be sure re-organize table 1 to include the individual age range of each person.
It is not possible to provide an individual age range for each person. Instead, we provided a global age range for each group. This has been clarified by a note in the legend of Table 1.

2. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only. This has been done accordingly.